Author’s response to reviews

Title: Nursing students' perception of clinical learning environment and supervision in relation to two different supervision model - a comparative cross-sectional study

Authors:

Mirjam Ekstedt (mirjam.ekstedt@ki.se)
Marlene Lindblad (marlene.lindblad@esh.se)
Anna Löfmark (anna.lofmark@hig.se)

Version: 4 Date: 09 Oct 2019

Author's response to reviews:

Dear Robin L. Cassady-Cain,

editorial office

We have responded to the request and sent the manuscript to a native editor for revision.

We also have checked that the manuscript conforms to the journal style and added a list of abbreviations.

The Royal Institute of Technology, KTH have changed the names of the institutions so we have also updated the institution at KTH on the title page.

We thus hope that we have completed your requests sufficiently.

on behalf of all authors

//Mirjam Ekstedt